Author’s response to reviews

Title: Personalized objects can optimize the diagnosis of EMCS in the assessment of functional object use in the CRS-R: a double blind, randomized clinical trial

Authors:

Yuxiao Sun (871743253@qq.com)

Haibo Di (dihaibo19@aliyun.com)

Lizette Heine (lheine@ulg.ac.be)

Wangshan Huang (944939037@qq.com)

Nantu Hu (nthu@psy.ecnu.edu.cn)

Xiaohua Hu (104216321@qq.com)

Xiaohui Fang (1141524042@qq.com)

Supeng Huang (385186187@qq.com)

Steven Laureys (steven.laureys@ulg.ac.be)

Haibo Di (dihaibo19@aliyun.com)

Version: 1 Date: 14 Oct 2017

Author’s response to reviews:

Dear Editors:

We are so sorry to make a mistake in authorship. Now we want to correct the order of authors and corresponding author details:

Yuxiao Sun, Jianan Wang, Lizette Heine, Wangshan Huang, Nantu Hu, Xiaohua Hu, Xiaohui Fang, Supeng Huang, Steven Laureys, Haibo Di
Thank you and best regards.

Yours sincerely,